Author’s response to reviews

Title: Restrictions Impeding Web-based Courses: a survey of publisher’s variation in authorising access to high quality on-line literature

Authors:

Dr Michele Langlois Dr (Michele.Langlois@man.ac.uk)
Prof Richard F Heller MD FRCP (Dick.Heller@man.ac.uk)
Richard Edwards MRCP MFPHM (Richard.Edwards@man.ac.uk)
Georgios Lyratzopoulos MRCP MFPHM (Georgios.Lyratzopoulos@man.ac.uk)
Dr John Sandars MRCP FRCGP (John.Sandars@man.ac.uk)

Version: 2 Date: 23 Oct 2003

As requested from the editors the revisions are to alter the format from a discussion article to a research article.